# REHABILITATION OF COVID-19 PATIENTS WITH ACUTE LOWER EXTREMITY ISCHAEMIA AND AMPUTATION

Luigia BRUGLIERA, MD, PhD<sup>1</sup>, Alfio SPINA, MD<sup>2</sup>, Paola CASTELLAZZI, MD<sup>1</sup>, Paolo CIMINO, MD, Pietro ARCURI, MD<sup>3</sup>, Maria Grazia DERIU, MD<sup>1</sup>, Chiara ZANETTA, MD<sup>4</sup>, Sara Monica ANGELONE, MD<sup>5</sup>, Jody Filippo CAPITANIO, MD<sup>2</sup>, Federica ALEMANNO, PhD<sup>1</sup>, Carlo MELONI, MD<sup>1</sup>, Giuseppe D'ANGELO, MD<sup>6</sup>, Elise HOUDAYER, PhD<sup>1</sup>, Jubin ABUTALEBI, MD, PhD<sup>1</sup>, Pietro MORTINI, MD<sup>2</sup> and Sandro IANNACCONE, MD<sup>1</sup>

From the <sup>1</sup>Department of Rehabilitation and Functional Recovery, <sup>2</sup>Department of Neurosurgery and Gamma Knife Radiosurgery, <sup>4</sup>Neurology Unit, <sup>5</sup>Clinical Psychology and Psychotherapy Unit, and <sup>6</sup>Department of Cardiac Electrophysiology and Arrhythmology, IRCCS San Raffaele Scientific Institute, Vita-Salute University and <sup>3</sup>Postgraduate Specialization in Physical and Rehabilitation Medicine, University of Milan, Milan, Italy

**Objective:** To evaluate the clinical characteristics and rehabilitation management of patients who undergo amputation for COVID-19-associated coagulopathy.

*Methods:* Clinical and laboratory data for 3 patients were analysed and their rehabilitative management discussed.

*Results:* The medical records of 3 patients who had undergone amputation due to acute lower extremity ischaemia and who were provided with rehabilitation in the COVID-19 unit at San Raffaele Scientific Insitute, Milan Italy were reviewed.

*Conclusion:* Coagulation changes related to SARS-CoV-2 may complicate recovery from this devastating disease. The rehabilitation management of amputated patients for COVID-19 acute lower extremity ischaemia is based on a multilevel approach for clinical, functional, nutritional and neuropsychological needs. Based on this limited experience, a dedicated programme for this specific group of patients seems advantageous to warrant the best functional outcome and quality of life.

*Key words:* COVID-19; SARS-CoV-2; amputation; COVID-19 coagulopathy; rehabilitation; ischaemia.

Accepted Jul 2, 2020; Epub ahead of Jul 16, 2020

J Rehabil Med 2020; 52: jrm00094

Correspondence address: Luigia Brugliera, Department of Rehabilitation and Functional Recovery, IRCCS San Raffaele Scientific Institute, Vita-Salute University, Via Olgettina 60, IT-20132, Milan, Italy. E-mail: brugliera.luigia@hsr.it

In Italy, severe acute respiratory syndrome coronavirus 2 (SARS-CoV-2) has resulted in more than 230,000 positive cases, with approximately 35,000 deaths, predominately in the north of the country. Recent studies have modified the diagnostic and therapeutic approach that was taken at the start of the pandemic, particularly regarding the target organs of coronavirus disease 2019 (COVID-19). These findings have also highlighted the need for a multidisciplinary rehabilitative approach to these patients in the acute phase, as well as in the postacute phase (1). Respiratory distress syndrome is not the only aspect of COVID-19; several other organs are directly affected, such as the heart, kidney, brain, and

# LAY ABSTRACT

COVID-19 is a devastating disease, which, in addition to pulmonary manifestations, causes several syndromes. Alteration of the coagulation system leads to thrombotic sequelae, which are typically associated with severe manifestations of SARS-CoV-2. These disorders can cause acute ischaemia of the lower limbs, necessitating amputation. This study reports the rehabilitation needs of 3 patients, highlighting the importance of a multidisciplinary tailored approach. Correct evaluation of the needs of this specific group of patients is essential to improve their chances of the optimum functional outcome and quality of life.

blood vessels, with endothelial damage, disepithelization, vasculitis, and development of micro-thrombi (1).

Patients with severe manifestation of COVID-19 have been shown to have important alteration of the coagulation cascade, such as disseminated intravascular coagulopathy (DIC) and thrombotic microangiopathy. The most common laboratory anomalies affect values of D-dimer, thrombocytopaenia, and prologned clotting time indicative for DIC, which differs from the sepsis-related DIC as it has high values of D-dimer and severe thrombocytopaenia. High levels of lactate dehydrogenase (LDH) and ferritin have also been reported, whereas autoptic surveys have showed platelet thrombotic deposits in the small vessels of the lungs and other organs (1). The physiopathology of the microvascular damage, thrombosis, and multiorgan failure caused by COVID-19 infection appears to be related to the affinity of SARS-CoV-2 for the angiotensin 2 receptor, which is thought to be the viral entry point into human cells, and is expressed in type 2 alveolar epithelial cells and endothelial cells (1-6).

So-called COVID-19-associated coagulopathy (CAC) is the array of coagulation alterations typical of infected patients (2–4, 7–10). This reflects a high prothrombotic status, which is associated with severe manifestation of the disease and increased mortality (6, 8).

Acute lower extremities ischaemia (ALI) is a severe consequence of hypercoagulability, the incidence of which is increased in patients with COVID-19 (2, 3, 9, 11). Moreover, in these cases a high incidence of

#### p. 2 of 4 L. Brugliera et al.

revascularization failure has been reported, probably due to excessive virus-related hypercoagulability (7, 8). The inefficacy of surgical and medical revascularization therapies and the clinical status of patients with COVID-19 leads to amputation of the affected limb. Once the patient has undergone amputation, a dedicated rehabilitation programme is then fundamental, which must take into account different and specific aspects, such as the patient's recovery after the acute phase of the infection and surgical intervention, management of pain, and preparation for the prosthesis prescription, manufacturing, fitting, and training. These aspects are all mandatory for the functional recovery of these patients and improvement in their quality of life (QoL).

The aim of this study was to review the clinical and rehabilitative findings from 3 cases of amputated COVID-19 patients who were managed in our unit, and to emphasize the importance of early rehabilitative management of these patients.

#### MATERIAL AND METHODS

The medical history of a series of cases that underwent rehabilitation in the COVID-19 rehabilitation unit in San Raffaele Scientific Institute, Milan, Italy were retrospectively reviewed, and their clinical and rehabilitation history discussed.

All procedures involving human participants were performed in accordance with the ethics standards of the institutional research committee and with the Declaration of Helsinki 1964 and its later amendments or comparable ethics standards. Informed consent was obtained from all participants included in the study.

## RESULTS

Three patients with amputation, who underwent rehabilitation in our department since the month of April 2020, were included.

### Case 1

A 58-year-old man with a history of hypertension, diabetes mellitus and hypothyroidism, underwent intubation for SARS-CoV-2. He was given hydroxychloroquine (HCQ) therapy. The patient was extubated due to improvement in his respiratory parameters after 7 days. A few days after extubation, the patient experienced progressive bilateral lower limb critical ischaemia, with necrosis of the left foot and microischaemic lesions of the right toes. Laboratory testing revealed: fibrinogen 718 mg/dl; LDH 738 U/l; and D-dimer >20 µg/ml. Platelet count, international normalized ratio (INR) and partial thromboplastin time (PTT) were normal. The patient underwent left lower limb transfemoral amputation, whereas medical therapy with low-molecular-weight heparin (LMWH),

aspirin, and vasodilator agents was started for the right lower limb. The patient was then admitted to the rehabilitation unit. Therapy was started for neuropathic pain, swallowing evaluation, neurocognitive and nutritional support, together with setting an appropriate protocol for a sacral pressure injury. The residual limb was then manually prepared for prosthetic positioning and a rehabilitation programme for the improvement of functional independence was started.

# Case 2

A 75-year-old man with a history of hypertension, hypertensive cardiomyopathy, diabetes mellitus, chronic kidney failure and peripheral neuropathy was admitted to the emergency unit for pain and hypothermia of the right lower limb. He had tachycardia, tachypnoea, hypoxaemia, and hypertension. Thoracic computed tomography (CT) scan revealed parenchymal thick-ening with a crazy paving pattern. Swab testing was positive for SARS-CoV-2. Laboratory testing revealed: fibrinogen 529 mg/dl; LDH 332 U/l. Platelet count, INR and PTT were normal.

The patient underwent antibiotic and antiviral therapy. Azithromycin (AZM); lopinavir/ritornavir, HCQ, and continuous positive airway pressure (CPAP) ventilation were started. Because of the patient's serious clinical situation and the damage to the microcirculation, no revascularization intervention was indicated and medical therapy with LMWH, aspirin, and vasodilator agents was started. Once the respiratory parameter improved, the patient underwent right transfemoral lower limb amputation with application of negative-pressure wound therapy. A few days after surgery, the patient was admitted to our rehabilitation unit. A specific rehabilitation protocol was started, to improve verticalization, postural changes and to prepare the stump for prosthetic fitting. Vacuum Assisted Closure® (V.A.C.®) therapy was continued for a small dehiscence of the surgical wound. The patients underwent swallowing evaluation, and nutritional and neuropsychological support. Neuropathic pain therapy was adjusted, and the VAC therapy was removed 10 days after admission to our unit. The patient is currently able to walk independently with a walker, and a customized temporary prosthesis is in production.

## Case 3

A 70-year-old man with history of hypertension, dyslipidaemia and chronic obstructive bronchopneumonia was admitted in the emergency unit for hyperthermia and dyspnoea. COVID-19 was detected on swab testing and a chest X-ray was consistent with pneumonia. AZM, HCQ, LMWH were started, together with CPAP.

Table I. Clinical and laboratory data for 3 patients amputated due to acute lower extremities ischaemia caused by COVID-19 infection

Case number	Age, years/sex	Affected lower limb	Fibrinogen (mg/dl)	D-dimer (µg/ml)	LDH (U/I)	Ventilation type	Medical therapy	ALI medical treatment	Amputation
1	59/M	Bilateral	718	>20	738	Intubation	HCQ	Iloprost, statin, ASA, LMWH	Transfemoral (left)
2	75/M	Right	529	N/A	332	CPAP	AZM, HCQ, LPV/r	Alpostradil, LMWH	Transfemoral
3	70/M	Right	152	5.04	711	CPAP	AZ, HCQ, TCZ	ASA, LMWH	Transfemoral

LDH: lactate dehydrogenase; ALI: acute lower extremities ischaemia; CPAP: continuous positive airway pressure; AZM: azithromycin; HCQ: hydroxychloroquine; LPV/r: lopinavir/ritonavir; ASA: acetylsalicylic acid; TCZ: tocilizubam; M: male; LMWH: low molecular weight heparin.

Due to worsening respiratory parameters, tocilizumab administration was started. Five days after hospital admission, ALI was detected, with thrombosis of the vascular axis of the right lower limb, and occlusion of the superficial, deep and common femoral and popliteal arteries. Laboratory testing showed the following values: fibrinogen 152 mg/dl; LDH 711 U/l; D-dimer 5.04 µg/ml. Platelet count, INR and PTT were normal.

The patient underwent emergent endovascular embolectomy with a Fogarty catheter. Because of revascularization failure, the patient underwent transfemoral amputation of the right lower limb 2 days later. A deep venous thrombosis of the left lower limb was detected one day later. One week after surgery, the patient was admitted to our rehabilitation unit. Following multidisciplinary evaluation, the patients commenced early rehabilitation, with postural training, manual management of the stump, and a silicon sheath. The rehabilitation programme was stopped after 7 days because the patient needed surgical revision of the wound and developed pneumonia. After 10 days, the patient started a new programme and is currently improving in autonomy.

In all 3 cases reported here the patients had CO-VID-19 respiratory syndrome. In one case intubation was necessary, and in the other 2 cases CPAP was administered. Antibiotics were administered in 2 cases (AZM; cases 2 and 3), antiviral treatment in 1 case, immunomodulant therapy in 1 case, and in all 3 cases the patients underwent HCQ. All laboratory testing showed a hypercoagulability status, which led to amputation at the thigh. Table I summarizes the clinical and laboratory data for this case series.

# DISCUSSION

SARS-CoV-2 directly affects several organs, leading to high morbidity and mortality (4). One of the main causes of morbidity is CAC, which seems to be the main cause of ALI (5). Hypercoagulability was found in all 3 cases reported here, as demonstrated by the levels of LDH, fibrinogen and D-dimer (4). Frequent amputation of patients with COVID-19 who develop ALI have also been reported, because of the higher rate of failure of revascularization techniques or medical therapies compared with non-COVID-19 cases (2, 4, 5). The rehabilitation management of COVID-19 patients with ALI is essential to improve functional autonomy and QoL. Stump management, with frequent dressing, manual mobilization treatment and analgesic therapy adjustments, are important for reconditioning the musculoskeletal system, which is compromised by the infection.

A tailored therapeutic-rehabilitation programme for these patients should focus on the recovery of functional autonomy, the disability, and on returning the patient as much as possible to social activities, in order to provide an acceptable QoL (12). A dedicated team of physiotherapists in our unit is involved in providing the individual rehabilitation programme, through maintenance of the elasticity of the residual limb, and prevention of post-operative contractures through selective training of the hamstrings, quadriceps, abductors, and adductors in the case of thigh amputation. Moreover, muscle-strengthening exercises, perceptive and coordination motor training, together with aerobic training for cardiorespiratory reconditioning and balance recovery are necessary (13).

Patients are also trained to adapt to the postural changes and recovery of verticalization. Finally, the correct mobilization aid is identified and the patient is trained in its use for the resumption of walking. In addition to physical management, patients in our unit are provided with psychological support and cognitivebehavioural therapy, which is known to reduce the anxiety and depression of this group of patients (14). Finally, amputated COVID-19 patients undergo continuous evaluation of their nutritional status, because, in optimal conditions, this improves the overall clinical status of these patients (15). Furthermore, optimal nutritional status is essential for the healing process and to prevent infections, especially for amputated patients with alterations in their microcirculation. Malnourished patients frequently develop pressure ulcers, infections, and delayed wound healing, which reduce the degree of clinical recovery and complicate the prosthesis of the stump (15-17). In our unit, we created an ad-hoc programme for amputated patients to start the processes of rehabilitation and prosthesis fitting. In most cases we positioned a postoperative silicone sheath, and thereafter a new system with a silicone suspension cap, once the stump was ready for the prosthesis (18).

#### p. 4 of 4 L. Brugliera et al.

In the early stage after admission to the rehabilitation unit, daily measurement is taken of the stump, 4 cm from the edge of the wound, and the patient wears a 3-mm silicone sheath instead of a conventional elastic bandage, to avoid the development of oedema and bottlenecks. The sheath is positioned for 1 h twice a day and progressively increased to 14 h a day. Once the stump is ready, usually from 7-10 days after surgery, a new system with silicone suspension cuff is applied, which differs from classic rigid cuffs for the flexible cup, allowing distal support. After delivery of the temporary prosthesis, 5–7 days after admission, it is possible to start prosthetic rehabilitation. After discharge from the unit, when the definitive prosthesis is available, the patient will again be admitted to the rehabilitation unit to start a new rehabilitation programme for at least 2 weeks, and thereafter the patient is followed in outpatient basis with the same medical and physiotherapy team to enable continuity of care. Even after patients have become independent in the management of their prosthesis and in activities of daily living, they are followed up every 2 months for 2 years, in order to address any new requirements or changes in their clinical status. Our previous experience has demonstrated considerable satisfaction among amputated patients, due to the advantages of early rehabilitation and use of a prosthesis. The same protocol was applied to patients amputated as a result of COVID-19. Future studies are needed to outline the correct rehabilitation programme for amputated COVID-19 patients.

## Conclusion

Coagulation changes related to SARS-CoV-2 result from a series of thrombotic changes that complicate recovery from this devastating disease. The rehabilitation management of patients amputated due to COVID-19 ALI is based on a multidisciplinary approach to the clinical and functional needs of disability. Neuropsychological and nutritional support is important for early rehabilitation, enabling functional recovery and adequate stump management. A dedicated programme for this specific group of patients is advantageous to warrant the best functional outcome and QoL.

The authors have no conflicts of interest to declare.

#### REFERENCES

- Brugliera L, Spina A, Castellazzi P, Cimino P, Tettamanti A, Houdayer E, et al. Rehabilitation of COVID-19 patients. J Rehabil Med 2020; 52: jrm00046.
- Bellosta R, Luzzani L, Natalini G, Pegorer MA, Attisani L, Cossu LG, et al. Acute limb ischemia in patients with COVID-19 pneumonia. J Vasc Surg 2020 Apr 29. [Epub ahead of print].
- Connors JM, Levy JH. COVID-19 and its implications for thrombosis and anticoagulation. Blood 2020; 135: 2033–2040.
- Sardu C, Gambardella J, Morelli MB, Wang X, Marfella R, Santulli G. Hypertension, thrombosis, kidney failure, and diabetes: is COVID-19 an endothelial disease? a comprehensive evaluation of clinical and basic evidence. J Clin Med 2020; 9: e1417.
- Tang N, Li D, Wang X, Sun Z. Abnormal coagulation parameters are associated with poor prognosis in patients with novel coronavirus pneumonia. J Thromb Haemost 2020; 18: 844–847.
- Walls AC, Park YJ, Tortorici MA, Wall A, McGuire AT, Veesler D. Structure, function, and antigenicity of the SARS-CoV-2 spike glycoprotein. Cell 2020; 181: 281–292 e286.
- Huang C, Wang Y, Li X, Ren L, Zhao J, Hu Y, et al. Clinical features of patients infected with 2019 novel coronavirus in Wuhan, China. Lancet 2020; 395: 497–506.
- Levi M, Thachil J, Iba T, Levy JH. Coagulation abnormalities and thrombosis in patients with COVID-19. Lancet Haematol 2020; 7: e438–e440.
- 9. Mucha SR, Dugar S, McCrae K, Joseph DE, Bartholomew J, Sacha G, et al. Coagulopathy in COVID-19. Cleve Clin J Med 2020 Jun 10. [Epub ahead of print].
- Zhang Y, Xiao M, Zhang S, Xia P, Cao W, Jiang W, et al. Coagulopathy and antiphospholipid antibodies in patients with COVID-19. N Engl J Med 2020; 382: e38.
- Baril DT, Ghosh K, Rosen AB. Trends in the incidence, treatment, and outcomes of acute lower extremity ischemia in the United States Medicare population. J Vasc Surg 2014; 60: 669–677 e662.
- 12. Sheehy LM. Considerations for postacute rehabilitation for survivors of COVID-19. JMIR Public Health Surveill 2020; 6: e19462.
- 13. Esquenazi A, DiGiacomo R. Rehabilitation after amputation. J Am Podiatr Med Assoc 2001; 91: 13–22.
- Bhuvaneswar CG, Epstein LA, Stern TA. Reactions to amputation: recognition and treatment. Prim Care Companion J Clin Psychiatry 2007; 9: 303–308.
- Brugliera L, Spina A, Castellazzi P, Cimino P, Arcuri P, Negro A, et al. Nutritional management of COVID-19 patients in a rehabilitation unit. Eur J Clin Nutr 2020; 74: 860–863.
- Carroll K, Baird JC, Binder K. Transfemoral prosthetic designs. In: Carroll K, Edelstein J, editors. Prosthetics and patients management: a comprehensive clinical approach. Thorofare, NJ: Slack Inc., 2006: p. 93–100.
- 17. Stechmiller JK. Understanding the role of nutrition and wound healing. Nutr Clin Pract 2010; 25: 61–68.
- Kern U, Altkemper B, Kohl M. Management of phantom pain with a textile, electromagnetically-acting stump liner: a randomized, double-blind, crossover study. J Pain Symptom Manage 2006; 32: 352–360.